Author's response to reviews

Title: Testicular Tuberculosis presenting with metastatic intracranial tuberculomas only: A case report.

Authors:

Godwin I Ogbole (gogbole@yahoo.com)
Oku S Bassey (tetebaby_tl@yahoo.com)
Clement A Okolo (clemabuokolo@yahoo.com)
Samson O Ukperi (uksamsonik@yahoo.com)
Ayotunde O Ogunseyinde (remioguns2001@yahoo.com)

Version: 4 Date: 11 October 2010

Author's response to reviews:

Dear Editor,

Re: MS: 1450003306372866_REVISION COVER LETTER

Thank you for asking for some revisions in the manuscript.

All the requested formatting changes have been made, addressing all the points raised.

1. A title page has been included as page 1, having the following: the title, authors, author affiliations and corresponding author details.

2. The ethnicity of the patient has been included in the abstract and case presentation section of the manuscript.

3. An authors contribution section has been included as suggested.

4. The header Case Report has been changed to Case Presentation.

5. We have done all to protect the anonymity of the patient by removing any identifiable information in the Case Presentation.

Thank you.